Author’s response to reviews

Title: Pain medication management of musculoskeletal conditions at first presentation in primary care: analysis of routinely collected medical record data

Authors:

Mehluli Ndlovu (mehluli.ndlovu@nottingham.ac.uk)
John Bedson (j.bedson@keele.ac.uk)
Peter W Jones (p.w.jones@keele.ac.uk)
Kelvin P Jordan (k.p.jordan@keele.ac.uk)

Version: 4 Date: 27 November 2014

Author’s response to reviews:

Author’s Response to Reviews

Title: Pain medication management of musculoskeletal conditions at first presentation in primary care: analysis of routinely collected medical record data

Authors

Mehluli Ndlovu mehluli.ndlovu@nottingham.ac.uk
John Bedson j.bedson@keele.ac.uk
Peter Jones p.w.jones@keele.ac.uk
Kelvin Jordan k.p.jordan@keele.ac.uk

Version: 4 Date: 27 November 2014

Comments from Associate Editor:

Please include all authors in the Authors’ contributions Section: Please include an Authors’ contributions section before the Acknowledgements and Reference list.

Authors’ response: We have amended the Authors’ contributions Section. It now reads (lines 404-406):

Authors’ contributions: MN, JB, and KJ designed the study. MN performed the analysis and led the writing of the paper. JB, PJ and KJ contributed to interpretation of results and contributed substantially to revising the paper. All authors approved the final version.